Run II Monthly Operations Report: Covering the period 2/20–3/19/06

Chronology
2/22: Store 4666 ended prematurely (1 hour in) due to quench. As a result of the quench a Tevatron magnet was damaged and it was decided to start the shutdown early (on 2/26) as a result. The cause: a failure in control card generated a bogus abort signal. Pbar losses on the abort caused a quench. A kautzky (relief) valve failed which was the immediate cause of the magnet damage. The November hardware failure was also a quench accompanied by a failed relief valve. As a result all Kautzky valves are now slated for maintenance during the shutdown. Plastic “popits” will be replaced on 1200 valves.
Shutdown Issues:

TEL-I appears to have suffered a vacuum rupture, probably as a result of the the 2/22 quench. Unclear how to repair.

Shutdown Activites and Status

We have completed the first five weeks of the fall 2004 shutdown. We are currently working to a 13 week schedule, with electron cooling installation in the Recycler as the critical path. Re-initiation of luminosity operations is scheduled for the week of December 6, providing us with two weeks for Tevatron startup.  Overall the shutdown is going well with the vast majority of activities on or close to schedule. The critical path of the shutdown, electron cooling installation in the Recycler, is slightly ahead of schedule. Current areas of concern include:

· The installation schedule for the Tevatron Ionization Profile Monitor, and its accompanying optical transition radiation monitor is very tight. We still hope to get both in during the shutdown.

Summarized below is the status of major activities being undertaken in the shutdown aimed at improved performance. There are myriad other activities that cannot be summarized here. Unless otherwise stated all listed activities are on schedule.

	Machine/Activity
	Goal
	Status

	
	
	

	Linac
	
	

	Drift tube replacement, tank 5
	Reduce losses.
	Awaiting cooldown (per schedule)

	
	
	

	Booster
	
	

	BRF19 (new rf station)
	Reduce losses during acceleration and provide added redundancy
	Cavity installed and being pumped down.

	L13 area reconfiguration
	Reduce losses during Booster acceleration cycle
	Awaiting cooldown. Starts next week.

	
	
	

	Antiproton Source
	
	

	Debuncher motorized stands
	Increase antiproton production yield
	10 (of 20) installed, 8 aligned

	Debuncher injection area rework.
	Increase antiproton production yield
	30% complete. ~1 week behind schedule but not critical path.

	AP1/AP2 Survey
	Increase antiproton production yield
	AP1 complete, starting AP2 this week.

	AP30 SO cord remediation
	OSHA violation
	Nearly complete (one week to go).

	
	
	

	Main Injector
	
	

	NuMI kicker installation
	Enable NuMI operations
	Kickers installed, awaiting cable connections

	Beamloading compensation
	Enable slip-stacking at 8E12 ppp
	Expect mid-October completion

	
	
	

	Recycler
	
	

	Stochastic cooling tank repairs
	Fix water leakes
	40% complete.

	Magnetic shielding in NuMI area
	Protect Recycler from NuMI beamline stray fields.
	95% complete.

	Electron cooling installation
	Enable another factor of>2 increase in luminosity
	Pelletron installation ahead of schedule. MI bus and LCW pipe relocation complete. Cable pulls complete. Stand installation complete. Overall on track for 13 week shutdown.

	
	
	

	Tevatron
	
	

	Install two new separators at D17
	Improved helix flexibility
	50% complete, ~1 week behind (but manageable)

	Cold mass (dumb bolt) shimming
	Re-center cold masses (reduce global coupling)
	122 (of 328) magnets shimmed. Ahead of schedule. May add another 86 magnets.

	Measure rolls on all magnets
	Update knowledge of Tevatron alignment
	Complete

	Unroll misaligned magnets
	Reduce global coupling
	On schedule

	Vaccuum upgrades at A0 and D0
	Improve operability
	On schedule

	
	
	

	NuMI
	
	

	MI extraction kickers
	NuMI extraction
	See MI.

	Extraction line magnets shielding
	Protect Recycler from NuMI stray fields
	See Recycler

	NuMI target hall installation
	NuMI operations
	~1 week behind. (Does not impact collider schedule.)

	
	
	

	MiniBoone
	
	

	Horn replacement
	Replace failed horn
	Cooling down. Failed horn will be replaced the week of 10/4.

	Infrastructure Maintenance
	
	

	Electrical distribution
	Preventative maintenance
	On schedule

	
	
	

	
	
	

	
	
	

	
	
	


